Author's response to reviews

Title: Age- and Gender-related Prevalence of Multimorbidity in Primary Care: the Swiss FIRE Project

Authors:

Alessandro Rizza (alessandro.rizza@eoc.ch)
Vladimir Kaplan (vladimir.kaplan@spital-muri.ch)
Oliver Senn (oliver.senn@usz.ch)
Thomas Rosemann (thomas.rosemann@usz.ch)
Heinz Bhend (heinz.bhend@sgam.ch)
Ryan Tandjung (ryan.tandjung@usz.ch)
The FIRE Study Group (heinz.bhend@sgam.ch)

Version: 5 Date: 28 September 2012

Author's response to reviews: see over
September 29, 2012

Journal Editorial Office
BioMed Central
BMC Family Medicine

Re: MS: 7771512297086762. “Age- and Gender-related Prevalence of Multimorbidity in Primary Care: the Swiss FIRE Project”

Dear Editorial Team,

I have rounded the proportions to whole numbers, in the abstract, as suggested. However, I have kept the use of double digits in the text part.

Thank you for publishing the manuscript.

Yours sincerely

Vladimir Kaplan, MD, MPH